\begin{table}%t8
\caption{\label{Si2-gf}A collection of the experimental and theoretical transition probabilities and Stark widths for the observed \ion{Si}{ii} lines. Errors are given in parenthesis.}
%\centerline
{
\begin{tabular}{lr|r|r|r|r|r|r|r|r|cc}
\hline
\hline
Wavelength & \exc &        & \multicolumn{7}{|c|}{\loggf} &\multicolumn{2}{|c}{$\log~\gamma_{\rm St}$}\\
\AA        & eV   &   NIST & Barach   &     SG   &    BBCB  &    BBC   &   Math   &    Wilke  &  AJPP&Wilke & LDA  \\
\hline   
3853.665 &  6.858 & --1.341 &          &--1.32(06) &--1.52     &--1.39(06) &--1.28(05) &--1.53(11)  &--1.44 &--5.12 &--5.15 \\%&2,3\\
3856.018 &  6.859 & --0.406 &          &--0.34(06) &--0.56     &--0.43(05) &--0.36(06) &--0.65(10)  &--0.49 &--5.15 &--5.15 \\%&2,3\\
3862.595 &  6.858 & --0.757 &          &--0.83(06) &--0.82     &--0.80(06) &--0.80(07) &--0.92(10)  &--0.75 &--5.14 &--5.15 \\%&2,3\\
4072.709 &  9.837 & --2.700 &          &--2.37(05) &	         &	         &--2.70(10) &--2.40(16)  &--1.89 &--4.89 &--4.76 \\%&3  \\%SG
4075.452 &  9.839 & --1.403 &          &--1.40(05) &          &          &--1.40(10) &--1.64(16)  &--0.95 &--4.89 &--4.76 \\%&3  \\
4076.780 &  9.837 & --1.700 &          &--1.67(05) &	         &	         &--1.70(10) &--1.75(16)  &--1.20 &--4.89 &--4.76 \\%&3  \\
4128.054 &  9.837 &  0.359 &          & 0.45(04) & 0.32     &          & 0.36(12) & 0.22(08)  & 0.38 &--4.92 &--4.88 \\%&3  \\
4130.872 &  9.839 & --0.783 &          &          &--0.82     &          &          &           &--0.77 &      &--4.88 \\%&3  \\
4130.894 &  9.839 &  0.552 &          & 0.50(04) & 0.48    	&         	& 0.55(12) & 0.38(08)  & 0.53 &--4.93 &--4.88 \\%&3  \\
4190.707 & 13.492 &        &          &          &	         &	         &--0.17(10) &--0.33(11)  &--0.35 &--5.25 &--5.26 \\%&no \\
4198.134 & 13.487 &        &          &          &	         &	         &--1.46(11) &--0.60(11)  &--0.61 &--5.26 &--5.26 \\%&no \\
4200.658 & 12.525 & --0.889 &          &          &	         &	         &          &           &      &      &--3.43 \\%&1  \\
4621.418 & 12.525 & --0.608 &          &          &	         &	         &          &           &      &      &--3.69 \\%&1  \\
4621.722 & 12.526 & --0.453 &          &          &	         &	         &          &           &      &      &--3.69 \\%&3  \\
5041.023 & 10.067 &  0.029 &          & 0.29(03) &          & 0.09(04) & 0.03(07) & 0.18(03)  & 0.19 &--4.84 &--4.78 \\%&3  \\
5055.983 & 10.074 &  0.523 &          & 0.59(03) &          & 0.46(04) & 0.52(08) & 0.48(03)  & 0.44 &--4.79 &--4.78 \\%&3  \\
5056.316 & 10.074 & --0.492 &          &--0.36(03) &          &--0.49(04) &          &           &--0.51 &--4.79 &--4.78 \\%&6  \\
5466.460 & 12.525 & --0.237 &          &          &	         &	         &          &           &      &      &--3.99 \\%&1  \\
5469.451 & 12.880 & --0.762 &          &          &	         &	         &          &           &      &      &      \\%&1  \\
5540.807 & 14.489 &        &--0.83(08) &          &          &          &--1.20(10) &           &--0.80 &      &      \\%&no \\
5576.661 & 14.504 &        &--0.47(08) &          &	         &	         &--0.47(10) &--0.62(13)  &--0.49 &--5.46 &      \\%&no \\
5632.966 & 14.186 & --0.818 &--0.62(08) &          &          &--0.82(06) &          &--0.82(09)  &--0.69 &--5.44 &      \\%&6  \\
5639.477 & 14.528 &        &--0.07(06) &          &          &--0.30(07) &--0.07(12) &--0.18(09)  &--0.32 &--5.44 &      \\%&no \\
5669.563 & 14.200 &  0.286 & 0.32(06) &          &          & 0.28(07) &          & 0.12(08)  & 0.25 &--5.53 &      \\%&6  \\
5688.817 & 14.186 &  0.126 & 0.08(06) &          &	         &	0.13(06) &          & 0.00(09)  & 0.08 &--5.50 &      \\%&6  \\
5701.370 & 14.175 & --0.057 &--0.10(06) &          &          &--0.06(07) &          &--0.28(09)  &--0.10 &--5.53 &      \\%&6  \\
5800.454 & 14.504 &        &--0.04(08) &          &          &--0.16(07) &--0.12(12) &--0.16(11)  &--0.17 &--5.37 &      \\%&no \\
5806.731 & 14.489 &        &--0.10(08) &          &          &--0.13(07) &--0.11(12) &--0.18(10)  &--0.20 &--5.47 &      \\%&no \\
5827.752 & 14.489 &        &--0.91(18) &          &          &--0.79(07) &--1.00(10) &--1.16(11)  &--0.90 &--5.45 &      \\%&no \\
5846.121 & 14.504 &        &--0.51(08) &          &          &--0.29(07) &--0.57(12) &--0.33(14)  &--0.70 &--5.35 &      \\%&no \\
5867.480 & 14.504 &        &          &          &	         &	         &          &           &--0.20 &      &      \\%&no \\
5868.443 & 14.528 &        & 0.42(06) &          &	         &	         & 0.40(14) & 0.50(11)  & 0.20 &--5.36 &      \\%&no \\
5957.559 & 10.067 & --0.225 &          &--0.30(02) &          &--0.22(03) &          &--0.29(06)  &--0.33 &--5.02 &--4.84 \\%&6  \\
5978.930 & 10.074 &  0.084 &          & 0.00(02) &          & 0.08(03) &          & 0.03(06)  &--0.03 &--5.01 &--4.84 \\%&2  \\
6347.109 &  8.121 &  0.149 &          & 0.18(05) & 0.30     & 0.20(04) & 0.12(07) & 0.29(08)  & 0.17 &--5.31 &--5.05 \\%&2  \\
6371.371 &  8.121 & --0.082 &          &--0.06(05) &--0.00     &--0.03(03) &--0.04(07) &--0.02(08)  &--0.13 &--5.32 &--5.05 \\%&2  \\
6660.532 & 14.504 &  0.162 &          &          &	         &	         &          & 0.23(09)  & 0.24 &--5.51 &      \\%&1  \\
6665.030 & 14.495 & --0.240 &          &          &	         &	         &          &--0.18(09)  &--0.16 &--5.54 &      \\%&1  \\
6671.841 & 14.528 &  0.409 &          &          &	         &	0.41(07) &          & 0.46(09)  & 0.52 &--5.58 &      \\%&6  \\
6699.431 & 14.495 & --0.247 &          &          &          &--0.25(08) &          &           &--0.16 &      &      \\%&6  \\
7848.816 & 12.525 &  0.316 &          &          &	         &	         &          &           & 0.34 &      &--4.34 \\%&1  \\
7849.618 & 12.526 & --0.831 &          &          &	         &	         &          &           &--0.80 &      &--4.34 \\%&1  \\
7849.722 & 12.526 &  0.470 &          &          &	         &	         &          &           & 0.50 &      &--4.34 \\%&1  \\
\hline   
\end{tabular}}
\smallskip
\par
NIST -- \citet{KP08} -- critical compilation; Barach -- \citet{Barach} -- pulsed arc; SG -- \citet{SG69} -- arc emission spectra;
BBCB -- \citet{BBCB} -- beam-foil lifetime measurements and theortical branching ratios; BBC -- \citet{BBC95} -- laser-induced plasma; 
Math -- \citet{Math01} -- laser-induced plasma; Wilke -- \citet{Wilke03} -- laser-induced plasma, Stark widths, shifts and transition probabilities;
AJPP -- \cite{AJPP81}  -- theoretical calculations; LDA -- \citet{Si2-Stark88} -- Stark widths, semi-classical calculations.
\end{table}